Author's response to reviews

Title: The size of the treatment effect: do patients and proxies agree?

Authors:

   Femke AH van der Linden (fah.vdlinden@vumc.nl)
   Jolijn J Kragt (jj.kragt@vumc.nl)
   Jeremy C Hobart (jeremy.hobart@pms.ac.uk)
   Martin Klein (m.klein@vumc.nl)
   Alan J Thompson (a.thompson@ion.ucl.ac.uk)
   Henk M van der Ploeg (vanderploeg-stapert@planet.nl)
   Chris H Polman (ch.polman@vumc.nl)
   Bernard MJ Uitdehaag (bmj.uitdehaag@vumc.nl)

Version: 2 Date: 17 July 2008

Author's response to reviews:

Dear editor,

as you requested I specified in the Results section that 7 of the 42 patients already participated in an earlier published study by BMC Neurology, including the reference.

Beste regards,

Femke van der Linden